Author’s response to reviews

Title: A randomised controlled trial of a physical activity and nutrition program targeting middle-aged adults at risk of metabolic syndrome in a disadvantaged rural community

Authors:

Krysten Blackford (k.blackford@curtin.edu.au)
Jonine Jancey (j.jancey@curtin.edu.au)
Andy H Lee (andy.lee@curtin.edu.au)
Anthony James (t.p.james@curtin.edu.au)
Peter Howat (p.howat@curtin.edu.au)
Andrew P Hills (ahills@mmri.mater.org.au)
Annie S Anderson (a.s.anderson@dundee.ac.uk)

Version: 3 Date: 31 May 2014

Author’s response to reviews:

Registration date added to manuscript.